Author's response to reviews

Title: Carotid intima-medial thickness measured on multiple ultrasound frames: Evaluation of a DICOM-based software system

Authors:

Kathleen Potter (pottek04@cyllene.uwa.edu.au)
Daniel J Green (d.j.green@ljmu.ac.uk)
Christopher J Reed (Christopher.Reed@health.wa.gov.au)
Richard F Woodman (Richard.Woodman@flinders.edu.au)
Gerald F Watts (gwatts@meddent.uwa.edu.au)
Brendan M McQuillan (Brendan.McQuillan@health.wa.gov.au)
Valerie Burke (vburke@meddent.uwa.edu.au)
Graeme J Hankey (gjhankey@cyllene.uwa.edu.au)
Leonard F Arnolda (Leonard.Arnolda@health.wa.gov.au)

Version: 2 Date: 27 August 2007

Author's response to reviews:

Dear Eugenio,

Please find included the figure legends that were missing in the original manuscript.

regards,
Kathleen Potter